Author's response to reviews

**Title:** The correct prednisone starting dose in polymyalgia rheumatica is related to body weight but not to disease severity

**Authors:**

Marco A Cimmino (cimmino@unige.it)
Massimiliano Parodi (parods@interfree.it)
Carlomaurizio Montecucco (montecucco@smatteo.pv.it)
Roberto Caporali (caporali@smatteo.pv.it)

**Version:** 5  **Date:** 21 April 2011

**Author's response to reviews:**

In the latest version the figures 2 and 3 have been cited in the text. They were erroneously indicated as tables.
The references have been put in square brackets.
The white part of the figures has been reduced.